Author’s response to reviews

Title: A cluster randomized controlled trial for child and parent weight management: children and parents randomized to the intervention group have correlated changes in adiposity

Authors:

Diane Berry (dberry@email.unc.edu)
Robert McMurray (exphys@live.unc.edu)
Todd Schwartz (Todd_Schwartz@unc.edu)
Emily Hall (eghall@email.unc.edu)
Madeline Neal (mneal@email.unc.edu)
Reuben Adatorwovor (radat223@email.unc.edu)

Version: 2 Date: 13 Nov 2017

Author’s response to reviews:

Dear Editor and Reviewers,

We thank you for your thoughtful review of our manuscript. We have made the following changes as requested.

1) Abstract: please provide the date of trial registration alongside the registration number. We have added the date as requested.

2) Ethical approval and consent to participate: please provide a statement to indicate that parents provided written informed consent. We have corrected this section as requested.

3) Availability of data and materials: it is not necessary to include 'There are currently several doctoral students using the data for their dissertations.' We have removed the comment regarding doctoral students using the data.

Thank you very much for the opportunity to respond to your concerns.